Author's response to reviews

Title: Duloxetine in treatment of refractory chronic tennis elbow: two case reports

Authors:

Zaid A Wani (zaidwani@gmail.com)
Shabir A Dhar (shabirdhar@yahoo.co.in)
Muhammad F Bhat (mbutt72@yahoo.com)
Yasir H Rather (yasirhrather@yahoo.com)
Shano Shiekh (shiekhshanoo@gmail.com)

Version: 5 Date: 28 July 2008

Author's response to reviews:

Dear Sir

As per the 2nd reviewers comments the following changes have been made in the manuscript.

1. “First report on the use of duloxetine to treat tennis elbow” added in the introduction.
2. Enhanced activity of nor epinephrine and serotonin in descending fibers by duloxetine could have resulted in the improvements observed in the two cases reported added in last paragraph of discussion.
3. Minor spelling mistakes correct in the manuscript and references.

All the changes have been highlighted by red ink

with regards

Dr Zaid A wani